Author's response to reviews

Title: Clinicopathological study of centrally necrotizing carcinoma of the breast

Authors:

Yanling Zhang (473444289@qq.com)
Yurong Ou (694699155@qq.com)
Donghong Yu (ydh_2210@126.com)
Xiang Yong (273513439@qq.com)
Xiaoli Wang (1558679005@qq.com)
Bo Zhu (35736791@qq.com)
Qiong Zhang (yongxiangkaoyan@163.com)
Lei Zhou (zhangyanling1204@163.com)
Zhaogen Cai (zyl70122@163.com)
Zenong Cheng (ydh2210@163.com)

Version: 8 Date: 25 March 2015

Author's response to reviews:

Dear Editor,

Good day!

Thank you for your comments. I have formatted and shortened the abstract. The Introduction has been re-named Background. Author full names have been changed into initials in Author's Contribution. Figures have be removed from the main manuscript.

If you have other questions, don't hesitate to contact me.

Thank you very much!